Author’s response to reviews

Title: External iliac vein compression and lower extremity swelling caused by an iliopectineal ganglion: a case report

Authors:

Kiyokazu Fukui (66406kf@kanazawa-med.ac.jp)
Ayumi Kaneuji (kaneuji@kanazawa-med.ac.jp)
Norio Kawahara (kawa@kanazawa-med.ac.jp)

Version: 1 Date: 15 Apr 2019

Author's response to reviews:

Editor-in-Chief, Journal of medical case reports

April 15, 2019

Point-by-Point Response to the Reviewers’ Comments

Re: Manuscript # JMCR-D-18-00803R1

Article title: “External iliac vein compression and lower extremity swelling caused by an iliopectineal ganglion: a case report”

Dear Editor:

We have revised our manuscript to address the reviewers’ comment. Our response follows comment below.

1. The Abstract is representative however please correct the dates the reflect the actual (i.e. 5 or 7 years).

Authors’ response:

Thank you for pointing it out. We have now revised the follow-up period correctly. The exact follow-up period was 7.5 years. Please see and lines 28, 53-54, 85 and figure legends of Fig. 4 and 5 in addition to abstract.
Note from authors

Please note that we have a bit changed the title. Also, please note that Fig. 5 has been revised a little.